Author’s response to reviews

Title: Impact of three different plate colours on short term satiety and energy intake: a randomised controlled trial

Authors:

Asli akyol (asli.akyol@hacettepe.edu.tr)
Aylin Ayaz (baylin@hacettepe.edu.tr)
Elif Inan-Eroglu (elif.inan@hacettepe.edu.tr)
Cansu Cetin (cetin.cansu@hacettepe.edu.tr)
Gulhan Samur (gsamur@hacettepe.edu.tr)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

Dear Reviewer,

Thanks for your valuable contributions. The corrections are listed below:

- Table 3: The legend is corrected

- As mentioned in line 82-83, white plate was used as control group and black plate was used to test the contrast difference with respect to red plate. Because of this reason, Bonferroni post-hoc test was used to detect any difference between black and red plates after observing the significant effect of main factor “plate colour”. Bonferroni post-hoc test is added in line 173-174, methods. As the main effect indicate the significant difference between red-white and black-white plates, authors kindly suggest to keep the current analyses.